Zayed University

ZU Scholars

All Works

6-10-2021

Internal and External Auditors Responsibilities and Relationships With Audit Committees in Two English Public Sector Settings

Karim Hegazy Zayed University, karim.hegazy@zu.ac.ae

Anne Stafford University of Manchester

Follow this and additional works at: https://zuscholars.zu.ac.ae/works



Part of the Business Commons, and the Social and Behavioral Sciences Commons

Recommended Citation

Hegazy, Karim and Stafford, Anne, "Internal and External Auditors Responsibilities and Relationships With Audit Committees in Two English Public Sector Settings" (2021). All Works. 4590. https://zuscholars.zu.ac.ae/works/4590

This Article is brought to you for free and open access by ZU Scholars. It has been accepted for inclusion in All Works by an authorized administrator of ZU Scholars. For more information, please contact scholars@zu.ac.ae.

INTERNAL AND EXTERNAL AUDITORS RESPONSIBILITIES AND RELATIONSHIPS WITH AUDIT COMMITTEES IN TWO ENGLISH PUBLIC SECTOR SETTINGS

Karim Hegazy *, Anne Stafford **

* Corresponding author, College of Business, Zayed University, Dubai, the UAE
Contact details: College of Business, Zayed University, Academy City, P.O. Box 19282 Dubai, the UAE
** Alliance Manchester Business School, The University of Manchester, the UK



How to cite this paper: Hegazy, K., & Stafford, A. (2021). Internal and external auditors responsibilities and relationships with audit committees in two English public sector settings [Special issue]. Corporate Ownership & Control, 18(3), 395–409.

https://doi.org/10.22495/cocv18i3siart13

Copyright © 2021 The Authors

This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0).

https://creativecommons.org/licenses/by/

ISSN Online: 1810-3057 ISSN Print: 1727-9232

Received: 16.03.2021 **Accepted:** 07.06.2021

JEL Classification: M4 DOI: 10.22495/cocv18i3siart13

Abstract

This paper assesses the job of scrutiny and oversight in public services by examining the role of the internal auditor (IA) and external auditor (EA) and their relationship with the audit committee (AC) in two distinct English public sector environments. The research uses an exploratory qualitative case study approach based on semi-structured interviews, AC meeting observations, and documentation reviews. The study acts as a starting point to examine the effectiveness of governance arrangements during a period of change in the UK public sector. The research provides evidence of good triangulation between the work of the IA, EA, and AC. Nevertheless, having close interaction between EAs and ACs and a crossing over in terms of responsibilities may lead to a conflict of interest and raises serious doubts about the independence and objectivity of the EA. This needs to be closely monitored over the coming years. Due to the diversity and wider and more complex accountability relationships and intricacies found in public sector organisations the study highlights the need to consider additional factors to the analysis of simple principal agency theory assumptions. The research further provides evidence on the existence of loosely coupled CG structures within the roles of the EA and IA within local government. This is one of the few papers which explores the IA and EA roles and their relationship with the AC in an organisational and institutional setting different from the private sector.

Keywords: Internal Auditor, External Auditor, Audit Committee, Agency Theory, Loose Coupling, English Public Sector

Authors' individual contribution: Conceptualization — K.H. and A.S.; Methodology — K.H. and A.S.; Validation — K.H. and A.S.; Writing — Original Draft — K.H. and A.S.; Writing — Review & Editing — K.H.; Visualization — K.H. and A.S.; Supervision — A.S.; Project Administration — K.H.

Declaration of conflicting interests: The Authors declare that there is no conflict of interest.

1. INTRODUCTION

This study examines the changing role of public sector oversight in the UK. It focuses on the role of internal and external auditors and their relationship with the audit committee in two distinct public

sector environments. First introduced by the National Audit Act in 1983, public sector auditing plays a major role in effective public sector governance which encompasses the policies, processes, and structures used by an organisation to direct and control its activities, to achieve its

objectives, and to protect the interests of its diverse stakeholder groups ethically (CIIA, 2020). To date public sector auditing has not been widely examined and "researchers in accounting have not been responsive to the problems and opportunities associated with developments in public sector accounting" (Hay & Cordery, 2018; O'Leary, 2015; Broadbent & Guthrie, 2008; Christopher, 2010; Goddard, 2010) even though spending on public sector services is immense¹. Furthermore, the public sector has complex processes, structures, and requirements where projects are often deemed more complex mainly providing explanations about the stewardship of public money and how it has been used in offering quality services, given the interest of the multiple stakeholders with legitimate various often conflicting but accountability expectations (Freeman & Evan, 1990; Ball, Grubnic, & Birchall, 2014). For instance, transparency and due process requirements applicable to the public sector exceed corporate requirements and all procedures and activities must be performed in a transparent manner. A further example of such complexity is the auditing of financial data which is extended in the public sector as value for money (VFM) auditing or monitoring of the three E's (economy, efficiency, and effectiveness).

In recent years, the UK public sector, in general, has seen significant changes as the new public management (NPM) ideology has driven a shift traditional public from a administration a "management culture, in the process emphasising the centrality of the 'customer' and accountability for results" (Aristovnik & Seljak, 2009). NPM elements include the use of private sector management styles, hands-on top management2, formal measurable standards and measurement of performance, and the unbundling of the public sector into corporatized business-like units and entities (Hood, 1991, 1995) geared towards pushing for managerialism rather than a more traditional public administration. An example of the effect of the NPM ideology, which also highlights the change in public administration, can be clearly seen with public sector audit. Changes have been evident in the public sector over the last few years (Van Schendelen, 2012), with an increased focus on the accountability, effectiveness, and efficiency of its services (Van Gansberghe, 2005). Citizens are demanding more transparency and better services, while resources are growing at a slower pace. This presents a unique set of risks for the public sector. As public sector organisations do not have the same built-in performance and accountability measures as the private sector (Jarrar & Schiuma, 2007) additional accountability structures, such as internal auditors (IAs), can add significant value. However, to address this need, public sector IAs should be sufficiently capable and efficient to provide the required level of assistance. Despite the increasing importance of IA, studies on the value-added of having an IA function in an institution, especially in the public sector, are limited due to the political and economic

environment of most public sectors and recent developments in the area (Brierley, El-Nafabi, & Gwilliam, 2001; Getie Mihret & Wondim Yismaw, 2007; Alzeban & Gwilliam, 2014). In 2015 the audit commission, which used to undertake the majority of external audits for public sector organisations, was disbanded by transferring the work to private sector companies. The aim was to bring the public sector audit regime into line with the private sector and mirror that of the private audit. Since 2015 the Public Sector Audit Appointments (PSAA) is responsible for appointing auditors to principal local government and police bodies, for setting audit fees, and for deciding the certification of housing benefit subsidy claims.

Thus, given the wider remit of the public sector and during these times of changes, the roles, and responsibilities of the audit committee (AC), seen as a central pillar of effective corporate governance (CG), have become increasingly demanding and scrutinized and calls for interactions with other relevant parties (FRC, 2018). While today's AC is seen as in the best position to provide effective oversight of the performance, independence, and objectivity of the external auditor (EA), a high-quality audit is also required. The AC is also considered an important tool for increasing the organisational status and independence of the IA function to provide internal audit oversight (CIIA, 2020). Although auditees may influence the IA's effectiveness, IA-EA relationships have largely gone underexamined in a public sector setting (Postula, Irodenko, & Dubel, 2020). Moreover, central regions and key organisational settings (e.g., not-for-profit organisations) have largely been absent in prior IA research (Kotb, Elbardan, & Halabi, 2020).

This study provides evidence on the role and job of scrutiny and oversight in public services by examining the role of the IA and EA and their relationship with the AC in two distinct public sector environments during the time of constant developments within the auditing arena (O'Leary, 2015). The study is also motivated by the paucity of studies that examine public sector governance structures and mechanisms in comparison to private sector CG research (Broadbent & Guthrie, 2008; Christopher, 2010). Thus, the research question can be formulated as follows:

What are the internal and external auditors' roles and responsibilities and their relationship with the audit committee in two distinct UK public sector environments?

This study takes a qualitative case study approach, based on semi-structured interviews, reviews, and documentation AC meeting observations, which enables information to be obtained from within organisations. This study extends previous work and makes an original contribution to the construction of new knowledge by providing evidence from two distinct public sector environments with differing histories, characteristics, and processes.

The paper finds a good triangulation within the roles of the IA, EA, and AC. Whilst the ACs often rely on the IAs' work, examining internal controls and risk management arrangements, the AC provides the IA with more credibility and teeth within the organisation. However, there were some contradicting findings in terms of the IAs appointment and the professional qualifications of

 $^{^1}$ In 2019 planned public expenditure in the UK according to the National Statistics (ONS, 2021) is £821 billion (£806 billion in 2018). As a comparison, the largest company in the world Walmart has revenues of £514 billion.

² Hands-on top management refers to managers who are highly involved in the day-to-day activities and decisions of the organisation to promote problem-free, productive operations.

its members, which raises serious concerns about the IAs' experience in LAs and whether they are fully equipped to fulfil their role efficiently. The study further finds that maintaining a good relationship between the EA and AC is significant as the AC draws on the advice of the EA given his wider financial experience and more flexible skills base. Another interesting finding is the closeness found between organisations and the EA especially in local authorities (LAs). This implies whether there will be a conflict of interest in the form of organisations having to pay the EAs fees after the abolition of commission, the audit which might the characteristics of independence and objectivity. This is also due to the newness of ACs especially in LAs who are still in a learning process and who are constantly being offered training on their roles and expectations by the EA. Furthermore, the findings confirm that having a good relation collaboration between the IA and EA leads to a more effective audit based on a clearer understanding of the respective audit roles and requirements and a better-informed dialogue with the AC on the risks facing the organisation.

The paper proceeds as follows. Section 2 reviews the relevant literature and theoretical framework used. Then Section 3 presents the main internal and external audit requirements found in two distinct English public sector environments. Section 4 presents the research methodology used. Section 5 presents the research findings followed by a discussion in Section 6 and conclusion in Section 7.

2. LITERATURE REVIEW AND THEORETICAL FRAMEWORK

According to the paradigm of corporate governance developed by the Institute of Internal Auditors (IIA), an effective IA is one of the four foundations of corporate governance along with management, the AC, and the external auditor (CIIA, 2020). The IA is an independent, objective assurance and consulting activity designed to add value which is critical for uncovering and preventing fraud in institutions (Abbott, Parker & Peter, 2010). The IA function has been made mandatory in private and public sector organisations due to the legal and regulatory reforms aimed at promoting better governance after the financial scandals of the early 2000s (Carcello et al., 2020; Sarens, De Beelde, & Everaert, 2009; Gramling, Maletta, Schneider, & Church, 2004).

CG guidelines and quotation rules (Blue Ribbon Committee, 1999; FRC, 2003, 2018) explicitly recognize the governance role played by ACs in supervising and supporting the relationship between EAs. management, IAs, and Subramaniam (2010) assert that in recent years, ACs have undertaken an important governance role in and monitoring coordinating the relationship between managers, IA, and EAs. Nevertheless, IAs and their relationship with the AC remains a neglected area of research (Gendron & Bédard, 2006; Roussy, 2013). Little has been written about the practice of the IA (Zain & Subramaniam, 2007; Davies, 2009; Roussy, 2013) and its relationship with the AC especially in an environment other than the private sector (Kotb et al., 2020), even though engaging in open and transparent relationships with the AC enhances the IA's effectiveness within the organisation (Beasley, Carcello, Hermansson, & Neal, 2009). Moreover, the AC is required to develop and implement a policy on the engagement of the EA to supply non-audit services, considering relevant ethical guidance by the EA. In addition, the AC is required to report to the board, identifying any matters in respect of which it considers that action or improvement is needed (FRC, 2018). Therefore, maintaining a good relationship with the EA is crucial for an effective working AC, as the AC also often relies on the EA's advice and expertise.

2.1. Internal auditors and audit committees

A considerable amount of literature has been published on different elements regarding the relationship of the IA and AC (Eulerich, Kremin, & Wood, 2019; Beasley et al., 2009; Sarens et al., 2013; Zaman & Sarens, Subramaniam, 2010; Lenz & Hahn, 2015). This is of importance as the AC can often act as a mediator and supporter of the IA in organisations (Turley & 2007) and has a role to assess the performance of the IA function, appoint heads of IA, support and promote the audit function within organisations (Davies, 2009), and strengthen the independence of the IA (CIIA, Consequently, the IA's familiarity in firms with risk management systems puts them in a unique position for providing an advisory, supportive, and facilitative role. Thus, they can offer great help and comfort to ACs to improve internal control and risk management aspects of companies (Sarens et al., 2009). In the same vein, O'Leary and Stewart (2007) shed light on the AC as an effective means of strengthening the position of the IA audit function by providing a supportive environment (Gramling et al., 2004) and Alzeban (2015) argues that the interaction of the IA function with the AC can establish a power base for the IA.

Research on the expectation and contributions of each other's role in the CG mosaic was also extensively studied (Sarens et al., 2009). These included the AC involvement in process issues (Gendron, Bédard, & Gosselin, 2004), the role of AC in negotiating resources for the IA function (Abbot et al., 2010), and the association between AC characteristics and disclosure and remediation of internal control weaknesses (Zhang, Zhou, & Zhou, 2007). These studies show that an IA function that receives strong support from the AC is likely to be more objective and powerful in the implementation of controls, thus allowing the IA to fulfil its role more sufficiently. The most discussed procedural however was found to be the meetings between the AC and IA (Gendron et al., 2004). ACs and IAs often engage in informal interactions in addition to formal pre-scheduled meetings. The establishment of an informal link between the AC and IA head was seen to improve the overall motivation of the IAs in carrying out their duties and their credibility to the AC chair as a valuable unit of the firm (Zaman & Sarens, 2013; Coetzee, van Rensburg, & Schmulian, 2016; Asiedu & Deffor, 2017).

Most studies though focused on the private sector. The limited prior public sector research has largely followed the patterns set out by researchers examining the relationships between all three parties in a private sector context, discussing IA public sector capabilities, their roles, composition, and effectiveness. Roussy (2013) found that the IA has two roles: a protector and helper role. On the one side, the IA has the role of protecting the top pitfalls or managers against any obstacles and keeping secrets besides supporting and the organisational performance providing guidance when new rules are implemented in a helper role. Thus, having a close relationship between the AC and IA improves the governance capabilities of both.

2.2. External auditors and audit committees

The AC relationship with the EA is not so close. being more formal (the Sarbanes-Oxley Act of 2002; FRC, 2018) than the AC relationship with the IA. Similarly, several studies have considered the role of the AC in appointing and removing the EA (Carcello & Neal, 2000, 2003; Krishnan, 2005). The AC often relies on the work done by the EA in terms of auditing and testing of controls. Regarding the AC and EA role, the AC has the "ultimate authority and responsibility to select, evaluate, and where appropriate, dismiss the outside auditor" (Blue Ribbon Committee, 1999, p. 14). Several associations between the AC, board characteristics, and auditor dismissal decisions (i.e., timing and successor auditor choice) have also been investigated (Chen & Zhou, 2007; Compernolle, 2018).

The AC is now also responsible for the appointment. compensation. oversight, retention, and dismissal of EAs. Auditor dismissals could occur as a response to the issuance of a qualified opinion, a going concern opinion, high audit fees, changes in client characteristics, and auditor effectiveness (Carcello & Neal, 2003; Hudaib & Cooke, 2005). Presumably, some dismissals are aimed at punishing auditors and/or seeking a more favourable audit opinion while others the objective of improving the audit service. It is clear nonetheless, that there is a rarity of studies examining the EA roles and their relationship with the AC in a public sector context. Especially, with the abolition of the audit commission, such relationship will increase in significance and there is a need to shed further light on their interactions from within organisations — as is the case in this study.

Thus, to recap, AC members depend upon both internal and external auditors in evaluating the effectiveness of internal control and financial reporting (Beasley et al., 2009). Therefore, having a good relationship with both is significant to the effective monitoring and challenging role ACs provide (Goodwin, 2004). However, the scant existing literature on examining such roles and relationships in a public sector context implies a need for further exploration research in different organisational and institutional contexts (Kotb et al., 2020).

2.3. Theoretical framework

Agency theory (AT) has been the overriding theory in CG studies and the predominant paradigm in studies concerning bureaucracy and public administration in the public sector (Schillemans & Busuioc, 2015; Gailmard, 2014). It has been used to explain issues relating to financial reporting and auditing in previous studies (Bradbury & Scott, 2015). In simple

terms, the AT portrays a contract between a principal (body or institution) who delegates powers to an agent to carry out the job. Agents, who often are assumed may operate in their best interest, are usually subject to the oversight and controls by the principal or other third parties (Bovens, 2007; Schillemans, 2013). IAs generally report to those who are strongly associated with "agents" (i.e., audit committee members) or to the agents themselves (e.g., top managers) (Burnett, Norman, & Sycara, 2011). Indeed, the IA, who is considered the first line of defence against fraud (Rezaee & Lander, 1993), must inform the AC about any irregularities or fraudulent activities of which he/she becomes aware. Conversely, IAs are more deeply submerged in organisational politics and are more often threatened by symbolic sanctions.

Thus, agency theory presents the foundation for explaining the functions assigned to IA since it is the IA that assures the management (Adams, 1994). Nevertheless, there has been a continued critique of its appropriateness in public sector institutions. The two main assumptions of AT, mainly the goal conflict between an agent and principal and information asymmetries, are said to vary as opposed to being constant, leading to the need to adjust such in a variety of institutional settings and organizational contexts (Meier & Krause, 2003). The behavioural assumptions of PA will not often hold as agents (public sector organisations) are guardians/stewards who are motivated to do their jobs and tied to an institutionalised mission that is likely to transcend their self-interest (Schillemans, 2013).

Thus, this paper argues that some of the initial assumptions of AT in institutional arrangements need to be revisited, relaxed, or dispensed of and extended explanations provided. There is a need to add additional factors to the analysis such as considering the wider and more complex accountability relationships and intricacies and auditing arrangements found in public sector various jurisdictions. Much of this demand has been because of public expectations in reaction to corporate scandals.

postulates that institutional theory The a corporation's management and arrangement tend to go with public expectations and has been powerful in explaining the adoption of innovations by institutionalised organisations and has been used extensively in public sector accounting (Cohen, Krishnamoorthy, & Wright, 2007, 2008; Ezzamel, Robson, Stapleton, & McLean, 2007; Jacobs, 2012) and how existing structures fulfil ritualistic roles to help legitimize the interaction among various participants of the organisation. For example. ACs may be coerced into becoming similar through regulation, following the best practice model, or by mimicking other organisations.

To be better equipped to understand and analyse the incentives and motivations that shape the different behaviour of agents the empirics in this study also consider the concept of decoupling as part of Institutional theory. Decoupling is one of the main coping devices by which organisations navigate complex institutional fields (Crilly, Zollo, & Hansen, 2012; Lok, 2010; Greenwood, Raynard, Kodeih, Micelotta, & Lounsbury, 2011). As a form of loose coupling, decoupling occurs when organisations adopt a legitimating program or policy

and at the same time fail to implement some or all the requisite practices expected to go the adoption. and it is typically used bv organisations as a means of reconciling conflicting institutional demands (Meyer & Rowan, 1977). Since organisations "face institutional complexity whenever they confront incompatible prescriptions from multiple institutional logics" (Greenwood et al., 2011, p. 318), decoupling serves as a particularly viable response in such institutional fields. Moreover, decoupling is widely considered to be response by which organisations purposefully "avoid" conforming to institutional pressures by hiding "nonconformity behind a facade of acquiescence" (Oliver, 1991, p. 154). As such, it fits well with the extant strategic understanding of institutional complexity: the conflicting demands in such environments are imposed upon organisations by various institutional constituents that adhere to different institutional logics.

In an auditing setting, this implies that the EA and IA roles and their relationship may only be loosely coupled with their effectiveness because their roles can be ceremonial/ritualistic and only there for external legitimacy seeking purposes (ceremonial conformity (Spira, 2002, 2003). Thus, the IA and EA roles and relationships may act to conform to pressures from stakeholders or triggered by power struggles (Modell, 2009a, 2009b), tend over time to become similar to others in the same industry or "absorb changes without necessarily changing their behaviour" (Marriott, Mellett, & Macniven, 2011). Some governance activities, processes, and structures may be driven by a desire to earn legitimacy, hence are ceremonial in serving as a symbol of effective oversight (Nor-Aziah & Scapens, 2007; Modell, 2009a, 2009b).

This study argues that the application of the theoretical assumptions of AT to the public sector is not sufficient. It agrees with the study by Lan and Heracleous (2010) who found that AT is only partially applicable to relationships between governments and not-for-profit organisations and that institutionalization and its different streams, would be more suitable to understand the adoption of CG processes and structures in public sector organisations. A single theory should not be expected to explain results in a study (Gendron, 2009) and in fact, diverse theories can be employed simultaneously to describe a given reality (Beasley et al., 2009; Cohen, Krishnamoorthy, & Wright, 2010).

3. BACKGROUND ON UK PUBLIC SECTOR AUDIT ENVIRONMENT

This study investigates the roles and relationships between the AC, IA, and EA in two distinct public sector environments, namely foundation trusts (FTs) and local authorities (LAs). The choice was made on the two largest public-sector areas with distinct histories and characteristics, and which have been undergoing significant changes especially in terms of public audit and governance. This would offer suitable evidence to examine such relationships.

Within health, the National Health Services (NHS) FTs are a relatively new organisational form and represent a good example of the application of NPM techniques to the public sector. FTs were created and set up as separate and independent legal entities. Through their centralized top-down

management structure, their governance model and structure are the closest in similarity to a private sector CG model. On the other hand, LAs have a long-established history dating back in some instances to Anglo-Saxon times. They, therefore, bring a wealth of tradition and piecemeal organisational structural change which makes it hard to impose governance new structures and regulations. LAs, which are political in nature, are created by Acts of Parliament and are set up as statutory autonomous bodies, independently elected, and can raise taxes locally.

3.1. Internal auditing

A professional, independent, and objective internal audit service is one of the key elements of good governance, as recognised throughout the UK public sector (CIPFA et al., 2017). The objective of the public sector internal audit standards is to set out the basic principles for carrying out the IA work and establish a framework for providing IA services in the UK public sector (CIPFA et al., 2017), and to have improved organisational processes operations to provide assurance on risk management arrangements, governance, financial and internal control. Boards should be advised by an independent AC chaired by non-executive directors (NED) and service operating in accordance the government IA standards.

The NHS internal audit standards (Department of Health, 2011) defines the nature of the IA within the NHS and sets the basic principles for carrying out the IA assurance and consulting role which includes giving an opinion on the overall adequacy and effectiveness of the organisation's framework of governance, risk management and internal controls in NHS organisations. These roles are required to be performed in accordance with requirements from the code of ethics which include integrity, objectivity, confidentiality, and competency, and mainly look at CG arrangements rather than performance. Moreover, the framework states, that the head of IA must establish effective communication with the AC chair with free and unfettered access and to inform the board on whether they have adequate assurances on the design and operation of the systems in place to mitigate any risks.

For local government, the IA is an ongoing function reporting to the council at least once a year and undertaken to test the continuing existence and adequacy of internal controls resulting in an assurance report designed to improve the effectiveness and efficiency of the activities and operating procedures of the council's internal controls (CIPFA et al., 2017). There is no requirement for the IA to be professionally qualified, but essential competencies to be sought in any IA include an understanding of accounting processes. It is however up to the LA to arrange for the IA either by appointment of an external IA firm or from in-house members. However, by choosing either option the appointment should comply with statutory requirements and IA standards, and ethical policies. It is beneficial for an IA to have awareness of risk management issues and an understanding of accounting requirements of the legal framework and powers of local councils (CIPFA, 2013). Moreover, it is essential that the IA function is sufficiently independent of the other

financial controls and procedures of the council which are the subject of review. To ensure the objectivity of IAs, it is now being recommended that the IA should hierarchically not be positioned underparts of the organisation that are themselves subject to internal audit scrutiny, thus have a reporting line that makes it independent of management.

3.2. External auditing

Public sector auditing takes a distinct approach and mainly focuses on VFM that can form a basis questioning those responsible and establishing future priorities besides assisting decision-makers in how to promote effectiveness (degree of objectives achieved) (Glynn, 1985; Parker & Guthrie, 1993). In 2015 the audit commission (set up 1983) was disbanded by transferring the work to private sector companies by outsourcing contracts aiming for more decentralization by organisations appointing their own auditors, increasing transparency to the public, ensuring high standards of the audit by effective and transparent regulation of public audit, and decreasing the audit fees by being more competitive. For the NHS, Monitor (now called NHS Improvement), the regulator has set out an audit code, which prescribes the way in which the EAs are to carry out their functions and reporting as set out in the National Health Service Act 2006. Besides giving an opinion on the financial position of organisations which implies the need to be independent, EA requirements in the NHS are more complex and geared towards reducing costs and burdens and looking at key areas of financial resilience. Moreover, it includes providing an opinion on whether organisations are performing effectively and responding to budget cuts during these times of austerity well, without affecting the quality of their services. In addition, auditing in the NHS includes a responsibility around quality accounts and giving opinion on whether there are arrangements for securing economy, efficiency, and effectiveness in the use of the available resources in line with their annual budget and medium-term financial strategy (National Audit Office & Office of Covernment Commerce, 2000).

For local government, the audit commission issued a separate code of audit practice, primarily to reflect the increasingly divergent accounting, CG, and performance management frameworks. Several principles also go further than private-sector codes such as reviews of compliance, reporting on the use of resources, VFM, certification, and audit reports (National Audit Office & Office of Covernment Commerce, 2000).

4. RESEARCH METHODOLOGY

This research is part of a larger study and draws on an exploratory case study approach which includes semi-structured interviews, document reviews, and observation of AC meetings. A qualitative approach was chosen for this study as it assists in understanding and investigating the roles of the IA and EA and their relationship with the AC by providing evidence and getting deep insights from within two public sector environments. These are beneficial when engaging with practice (Yin, 2003)

and offer better descriptions of the phenomenon because they permit details naturally suppressed in studies of large samples (Patton, 2002). One of the important aspects of qualitative and interpretive research is a well-kept balance between rigor and openness (Ahrens & Chapman, 2006). Openness can be achieved through methodological flexibility and multivocality, which, according to Gendron (2009), are not independent of one another. Methodological flexibility allows for the adoption of data collection and analysis according to the emergence of significant trends and patterns from the data.

As this study is part of a wider study and based on results of compliance testing performed, four FTs and four LAs reflecting both high and low levels of compliance were chosen for further investigation. This was later reduced to two FTs and two LAs based on the organisations' willingness to be interviewed, the research objectives, the type of the FT, the political structure of the LA, and the accessibility of the organisations chosen. Table 1 presents the main characteristics of the four cases chosen at the time of the study and for anonymization purposes, the FTs chosen were named North and South, while the LAs were named East and West. In total and for the purpose of this study 26 interviews were carried out in four public sector organisations between the years 2014-2016. Interviews were undertaken with finance directors, internal and external auditors, audit committee chair, and members of all cases. Due to privacy requirements in the health sector, the AC chairs were asked for permission to attend AC meetings on an observation-only basis. In contrast in local government, AC meetings were open to the public except for any restricted business of a confidential nature, and minutes and some working papers are made publicly available on websites.

Semi-structured interviews were predominantly used to gain insights from within organisations on the interaction between the AC, IA, and EA in practice and helped to uncover any non-visible data which may be due to limitations of secondary resources and to seek confirmation and clarity to increase data reliability (Griffee, 2012). Before the interviews took place, secondary data were primarily obtained from publicly available sources such as annual and governance reports, public service rules and guidelines, codes and frameworks, corporate websites, news and press items, and board minutes and agendas. These publicly available data were examined to generate a set of questions. including points about the role and responsibilities of the interviewee, follow-up, and confirmation of information coming out of previous board meetings via minutes and agendas to check whether they have been implemented or not. Furthermore, meeting observations provided supplementary insights and verifications of participants' claims as well as confirmation of documentary reviews (McKernan, 2013). Examining secondary data along with the interviews and meeting observations was used as part of a triangulation process to compare and interpret information and reveal any contrasting views between interviewees' responses and written records. Once the data had been collected, collated, and transcribed, they were manually coded using key theoretical constructs (Ahrens & Dent, 1998). Patterns and exceptions were identified in the coded data and the results were documented once the process was complete.

Table 1. Description of case studies

Parameter	North	South
Became FT	2006	2006
Board composition	Unitary board, chairman, CEO, 5 executives, 6 non-executives	Unitary board, 5 executives including CEO, 6 non-executives including chairman, CEO acting as accounting officer
Governors	12 elected and 5 appointed, 9 staff governors	20 elected representing public, 7 representing staff and 5 appointed, roles advisory and strategic direction
Committees	8 overall including audit, remuneration, nomination, terms of service, risk, and assurance	3 statutory committees, audit, remuneration, nomination, healthcare risk, and governance
EA and IA	EA (audit commission) IA (Merseys)	EA (Grant Thornton) IA (KPMG)
Risk rating by the regulator	Financial: 3 Governance: Amber Green	Financial: 3 Governance: Amber Green
EBITDA	23.2 million Deficit: 14.1 million Income: 397 million	22.1 million Surplus:2.68 million Income: 385 million
Income from patient care	80%	60%
Expenditure	£230 million Net assets 109120 million	£114 million Net assets 85923 million
Parameter	East	West
Organisational structure	96 councillors 9 members represent the executive team which includes officer for adults, environment, culture, and HR. Amongst others	60 councillors City Major, deputy plus strategic assistants. Cabinet of 10 with voting rights plus 5 assistant majors for finance, housing technology, etc.
Committees	6 overview and scrutiny committees', planning and development, and AC	11 committees excluding scrutiny, 6 scrutiny including children and health and budget, 8 community committees
Political structure	86 Labour, 9 Lib Dem, 1 independent	52 Labour, 8 Conservatives
AC composition	9 members, 6 from ruling party Labour, rest mixed	10 members, chair from an opposition party, 7 Labour, 2 independents
Revenue	538 million budget 75% from the government the rest from tax grants, business rates, etc.	228 million budget 41% coming from tax, 51% from business rates, and 16.5 from grants
External auditor	Grant Thornton	Audit commission
Internal auditor	Grant Thornton	Grant Thornton
Audit fees	Proposed fee: £493000	Scale fee: £182000

5. RESEARCH FINDINGS

5.1. AC relationship with the IA

In public sector organisations, the main purpose of the IA is to review whether the financial systems, governance standards and assurance, risk management arrangements, and financial controls and operating procedures are effective.

"The IA gives an opinion on the FTs systems of governance, risk management, and internal control but I see their broader role in the organisation managing itself better to have better governance, better control, and risk management" (FT member).

The findings further provide evidence of the cycle the IAs go through to present their work to the AC who in turn reviews and scrutinises their work. The IA usually reports through the management route first, which means that the personnel responsible from the management side (usually the finance director or equivalent) meets monthly with the head of IA to go through the progress of the IA plan and any emerging issues before meeting with the AC. However, the IA also has a special route to report up through the AC on what has and is being done and how VFM is being delivered. This link is of significance as it illustrates that the AC is kept up to date and provides the AC with constant assurances on happenings in lower levels of the organisation. The IA reports, which usually consist of set pieces and where the year-end document is a set of accounts, which pictures measures of the IA performance and comprises the IA's annual audit opinion on the internal controls of the organisation, are then presented in

a special AC meeting as head of IA on a yearly basis. The AC then in turn reviews the IA's work and monitors the progress against recommendations of reports (progress against IA plan or outcome from IA reports) provided quarterly by the IA. Moreover, the AC also must get a sign-off from the IA in terms of reasonable assurances that the internal controls are working effectively and that the major risks are within an acceptable level, to be able to certify the integrity of the accounts in the organisation. These findings confirm a good interaction between the work of the IA and AC and ensure there is the constant awareness of the ongoing work of the IA team by the AC, which is in line with best practice ideas and formal guidelines.

The research findings further confirm that there is a close, easy, relaxed, and good working relationship between the IA and AC as the IA is regarded as "the day-to-day eyes and ears of the operation of the trust". Although most of the IA work is done and reported first to management the IA can get in touch with the AC at any time.

"There is a quite easy way of communicating in that respect, so she knows she can always speak to me as chair. Likewise, I know that if I need to pop in this morning without an appointment, I can do so" (AC chair).

"I find the relationship to the IA extremely good because the IA understands the business and the work, he does have a whole range of stuff and a whole range of powers. IA is the most interesting part of the work on the AC" (AC member). "I have an incredibly positive relationship with

"I have an incredibly positive relationship with the chair, and I see it as particularly important. As I report independently, I can do that in a public committee that gives me a link through to the political side of the business" (IA member).

An FT AC chair further reinforces that he finds the IA people experienced and sensitive to the fact that this is a hospital (trust) whilst demanding the best in terms of The importance of the close AC and IA relationship also lies in the fact the combined work in terms of setting the audit plan. Therefore, it is of significance that the AC and IA maintain integrity and competency when dealing with and relying on each other's work. The findings however find that the appointment of the IA does not follow a certain pattern. In one instance the IAs were found to be from in-house, in another, they were externally appointed. Nevertheless, in both instances, they are bound by a code of ethics which includes maintaining integrity, objectivity, confidentiality, and competency when performing their role. The latter focuses more on management style issues, governance arrangements, and the internal control environment.

The research, similarly, to findings from the private sector (Gramling et al., 2004; Sarens et al., 2009), further provides evidence that the AC contributes to the role of the IA as it adds more credibility to their work and gives them "more teeth" in the organisation. The AC is there to challenge and scrutinize the work of the IA as well as being there to help them with any authority matters. "Often issues and what we say may not be taken notice of, so the AC can help by stepping in and adding weight to what we are saying" (IA member).

Alzeban (2015) supports this view by arguing that the interaction of the IA function with the AC can establish a power base for the IA with the AC involvement in process issues and in negotiating resources for the IA function to operate effectively and strengthen their credibility. However, important to stress is that the IAs do not consider themselves that they work for the AC: "Previously the AC would actually decide the IA work program which is not the way it should actually work" (IA member).

This quote illustrates that there are continuous developments within the understanding of the IA roles and their interaction with the AC in line with best-practice processes. On another the interviewees agree that AC members sometimes feel the need to be protected by the IA too as they are aware that they only have access to information filtered by managers of the organisation. However, in some instances, they may want additional insider information accessible only by the IA. Thus, the AC relies on the IA to warn them of issues that managers wish to conceal but also to direct the IA to focus on major risks facing the organisation. Consequently, for the AC to be able to fulfil its monitoring and scrutinizing role effectively it needs to have a clear and transparent relationship with the IA as both rely on each other's help and advice enabling them to perform their prescribed roles

Thus, maintaining a good relationship between the AC and IA is significant as they both often rely on the work of one another which improves the governance capabilities of both (Postula et al., 2020; Zaman & Sarens, 2013; O'Leary & Stewart, 2007; Roussy, 2013). Although the AC monitors and scrutinizes the work of the IA, the AC in its authoritative role provides the IA with more comfort

and credibility within FTs and LAs. Considering the symbiotic relationship between the AC and the IA function, their effectiveness and performance are inevitably interrelated (Sarens et al., 2009; Zaman & Sarens, 2013).

5.2. AC relationship with the EA

The interviewees confirm that the EA has a much wider remit to cover in public sector organisations. Unlike the private sector where EAs are required to give an opinion on whether the accounts show a true and fair view and confirm that the company is a going concern, auditing in the public sector goes beyond that.

"In the FT sector, there is responsibility around quality accounts which is quite specific. There is also a responsibility on giving an opinion on whether there are proper arrangements for securing economy, efficiency and effectiveness in the use of the available resources" (AC chair).

"In local government aspects as maintaining financial sustainability, waste and recycling targets, and maintaining budgets on expenditure on Children's and Education and Housing Services which all relate to improving VFM are aspects EA have to look at" (EA member).

These aspects are all of significance to the various stakeholders with legitimate vested interests such as members, patients, or the public who demand high-quality care and services and transparency in reporting.

The demise of the audit commission has led to appointment private the of new firms. The interviewees believe that having a balance on the EA team between old members (from the audit commission), given their knowledge and experience about the structures and processes the organisation and the new EA members (from private firms), with fresh and useful thoughts, are expected to be beneficial to the organisation. However, there will be requirements for rotation in a few years, as in the private sector, to comply with the independence and objective elements auditors possess. It remains to be seen whether private firms, with little experience of the structures and processes of public sector institutions, are well suited to take over the role of the audit commission, during the continuous changes in governance arrangements, the UK public sector is undergoing. The research findings found that unlike with the IA, the AC does not meet the EA as frequently because of the different natures of both their jobs. In public organisations, the AC relationship with the EA is about the latter being able to give the AC members assurance that the accounts having withstood scrutiny can meet all requirements of the regulator in terms of any accounting standards and reporting requirements.

The EA relationship with the AC is being seen by some interviewees, especially in the health sector as a detached and formal relationship and being more at arm's length. However, other interviewees especially in local government, view the relationship with the AC as an informal communication and relaxed relationship. "If they want to meet with us privately over and above the normal set periods then they can do so. And in that respect, there's an open door for them to do that" (EA member).

These contradicting findings raise thoughts on the different environments understanding of the interaction and relationship between the EA and AC. All interviewees nevertheless strongly believe that maintaining a good relationship between the AC and the EA is crucial to having sound CG structures.

"The code of governance would always say there needs to be time for auditors and the AC to meet privately. Out of the four AC meetings a year, that would happen on three occasions. It is always important that the chair of the AC and the audit manager have each other's numbers" (FT member).

A further AC chair sees the relationship with the EA as healthy too and confirms having a good working relationship with the EA. The EA can meet with the AC chair at least once a year outside the AC. The only criticism is that it can get "cosy" when the organisation itself starts paying the EA wages after the abolition of the audit commission. "According to the master-servant relationship," there should always be that distance between the council and the EA" (AC chair).

This point raises serious thoughts about the independence aspect of the EA which might hinder its role in terms of reporting in an objective and independent manner. More interestingly the research findings confirm such concern and find that in practice there is a crossing over in terms of responsibilities which affects the independence aspect of the EA and AC. So recently in an organisation the EA conducted a survey of the AC members' thoughts on their role to be used as a sounding board outside of the committee to help them with any necessary training required. "I feel that the AC members are seeing that we are bringing the AC some assurances. The dialogues and questioning are there and often there are queries on our reports and the AC members want a further understanding of it. So, the AC is there to critically assess whether we are doing our work in some aspects correctly" (EA member).

A further quote by an EA raises doubts about the interaction and closeness between the AC and EA and whether the private firms have considerable experience and knowledge "If they want to meet with us privately over and above the normal set periods then they can do so. And in that respect, there's an open door for them to do that" (EA member).

Furthermore, informal meetings often also take place to discuss relevant matters and concerns by the EA. On one hand, informal meetings are a good opportunity for the AC members to ask questions, whereas the EA could make the committee aware of any financial or non-financial issues and raise any concerns. It is encouraged as a form of standard practice for the EA to meet alone with the AC as it provides an important opportunity for the AC to raise issues, ask questions and seek feedback from the EA without the IA or management observers being present. One example of this happening in practice was observed during an AC meeting attended. An AC chair suggested he had an informal chat with the EA to discuss several issues regarding the new auditors' engagement as well as enquiring more about the audit approach the new private firm is planning to undertake in terms of their reporting style.

Having a relationship based on trust between the AC and EA enhances the reliability of the ACs monitoring and scrutinizing as the AC often relies on the EA's opinions, especially on the financial accounts. However, on the other hand, informal meetings are also often used for other purposes which suggest less independence and a crossing over in terms of AC and EA roles. As an example, the EA often holds presentations for the AC members to try and get them up to speed with any issues, emerging requirements, or regulations. This is due to the newness of the AC in public sector organisations, especially in local government. The findings further suggest that the AC are still in a learning process and are gradually improving in terms of their effectiveness and awareness.

The findings confirm a good working relationship between the AC and the EA in all organisations as they often build on each other's work for obtaining additional assurance. Nevertheless, the findings suggest a close interaction between the AC and EA and a crossing over of responsibilities, especially in LAs, which raises questions about the independence aspects of both. It remains to be seen how such a relationship develops in the future.

5.3. Relationship between the IA, EA, and AC

For a good relationship to exist between IAs and EAs, the independence of the IA department is critical as often EAs rely on the work of IAs to do their job. The scope of work and materiality of systems and structures depend on the efficiency and effectiveness of the IA. For instance, a poorly deployed internal auditing system can lead to increased, non-value-added costs, many hours of wasted resources, and eventual breakdown of the assurance system of the institution. The research findings confirm the close relationship between the IA and EA. The IA meets around four times with the EA in informal meetings between them besides also meeting in the AC meetings. The formal as well as the informal meetings between the IA and EA are significant as the IA and EA review and often place reliance on each other's work.

"I think it's important to give their opinion and they need to get an understanding of the accounts, VFM, and what else is going on in the business. I can help them understand this. Equally, I rely on them to do some work rather than duplicate our work. Overall, we have a constructive relationship as we are two parts of good governance arrangement" (IA member).

Usually, the IA is aware of aspects that go on at an organisational operational level, and thus strategically the IA head has a particularly good understanding of where the organisation is heading. On the contrary, similarly as with the AC, the EA usually only has and can gain access to information he receives from managers or officers. This means he often must rely on more insider information provided by the IA. On the other side, the EA can also advise the IA on certain areas where some form of external expertise may be needed, e.g., in terms of management controls. Furthermore, as most of the EA's work is around financial accounts.

³ An archaic generic legal phrase that is used to describe the relationship arising between an employer and an employee. A servant is anyone who works for another individual, the master, with or without pay. The master and servant relationship only arises when the tasks are performed by the servant under the direction and control of the master and are subject to the master's knowledge and consent.

If the IA can provide assurance that systems and arrangements operate effectively and there are no significant errors, then the EA can gain some comfort from that and help them with their audit approach. More importantly, having sound CG structures and internal controls in place benefits the AC in terms of its monitoring and scrutinizing role.

Furthermore, the IA and EA usually consult with each other during the planning phase of individual audits that address the key financial and business systems underpinning the organisations' accounts. To confirm with best practice arrangements, the EA is required to evaluate the work of the IA to determine its adequacy for external audit purposes (Prewitt & Erhun, 2011). However, there have been developments as the IAs now direct the EA too by requiring them to perform work on specific aspects they may need further assurances on. "It's not seen in a confrontational way but rather as more of a partnership. Nevertheless, the EA still possesses a key independent role to play in the organisation" (IA member).

This quote again raises concerns about the independence aspect of the EA and whether such close interaction is healthy to having sound CG arrangements within the organisation. Such informal relationships also surprised an AC chair: "I never quite understood, particularly before becoming a member of the AC how informal it is. I would have imagined that there should have been a stricter relationship between the EA and IA. But it does not really work that way. It is a very informal relationship and clearly, they have to work together a lot throughout the year".

The findings further confirm that there may be some elements of overlap within the IAs and EAs agenda and they also may work together on particular areas if needed. An IA confirms such a view but ensures that when any collaborative work is undertaken the work balance doesn't conflict: "The EA basically lives here as they have an own office at the council, and we see them daily. We have formal meetings with the CFO once every 6 weeks to talk about issues the EA may have or work, they must do".

Nevertheless, important to note is that such an informal relationship also has its benefit because if any sudden issues come along such as a discovery of fraud, the IA would inform the EA straight away rather than wait for the next meeting. This would enable the EA to tackle any evident issues and risks immediately.

6. DISCUSSION

Through the lens of agency theory, the IA and EA and their relationship with the AC are part of a governance arrangement in firms to satisfy the demands for accountability and bond the contractual relationship between stakeholders and agents. Whilst the ACs often rely on the IAs' work, which predominantly examines the internal controls and risk management arrangements of the organisation, for insider or management lower-level information, which is not easily accessible, the AC provides the IA with more credibility and teeth within the organisation. The informal relation between both and the IA having unfiltered access to the AC means that the AC is constantly kept up to date on any emerging matters which improves the scrutinizing and monitoring responsibilities. However, the AC must assess and review the IA's performance too, which requires maintaining a balance between being professional, independent, confirms objective. The study also the continuous developments within the IAs role and a better understanding of the IAs relationship with the AC. Historically the AC was responsible for setting the IA roles and plans which should not be the case.

There were some contradicting findings in terms of the IAs appointment and the professional qualifications of its members which contradicts some of the assumptions of AT. The flexibility in terms of appointing IAs from in-house or from outside of the organisation and having no requirement to be professionally qualified raises serious concerns about the IAs' experience in LAs and whether they are fully equipped to fulfil their role efficiently. Due to the constant changes in the public sector, there are elements of a loosely coupled system, as part of Institutional theory, evident in terms of absorbing changes to create legitimacy outside of the organisation.

Additionally, the study finds that maintaining a good relationship between the EA and AC is significant as the AC draws on the advice of the EA given his wider financial experience and more flexible skills base. Nevertheless, due to the wider EA responsibility in the public sector, which not only covers financials but also quality accounts in FTs and VFM in LAs, the question arises whether organisations have suitable and enough resources to cover such responsibilities. With the demise of the audit commission and the appointment of new private firms, there is a balance between old members given their knowledge and experience processes structures and about the the organisation and the new EA members, with their fresh and useful thoughts. This raises thoughts on whether these arrangements, in terms of allowing governance continuation of previous arrangements and personnel and just appearing to have absorbed such changes in personnel in a ceremonial way, are beneficial or whether there are elements of loosely coupled structures evident. The question also arises whether private firms, with little experience of the structures and processes of the public sector, are well suited to take over the audit commission the role of the public the continuous changes sector is witnessing.

Another interesting finding which needs to be observed over the coming years is the closeness found between organisations and the EA especially in LAs. This implies whether there will be a conflict of interest in the form of organisations having to the EAs fees, which might the characteristics of independence and objectivity thus contradicting AT assumptions. This is also due to the newness of ACs especially in LAs who are still in a learning process and who are constantly being offered training on their roles and expectations by the EA. Moreover, the relationship between the AC and EA was found to be more of a formal and detached relation at an arm's length in FTs, similar to private sector practices, whereas more of

an informal relationship in LAs. The issue arises about the different understanding found in both environments about the EAs and their responsibilities and raises serious questions about whether this might affect the EAs' objective monitoring and reporting. This point is further reinstated by the EAs being located and having their own office in a LA.

Furthermore, the findings confirm that having a good relation and collaboration between the IA and EA leads to a more effective audit based on a clearer understanding of the respective audit roles and requirements and a better-informed dialogue on the risks facing the organisation. The EA can often rely on the IA to obtain any insider information and the initial testing of the internal controls performed by the IA whereas the EA can also advise the IA on certain areas where some form of external expertise may be needed. This leads to the focusing of audit effort and consequently to more useful advice to the AC. Therefore, the interrelationship between the three parties (AC, IA, and EA) is crucial for the organisation in terms of having sound and proper governance arrangements and an effective working AC which enhance the organisation and assist senior management and the AC in providing high-quality public services.

7. CONCLUSION

This study provides insights into the role of public scrutiny and oversight and examines whether governance arrangements in terms of the IA and EA roles operate effectively in two distinct public sector environments. The research is exploratory in nature and has adopted a qualitative case study. It has done so through providing evidence from the two largest UK public sector environments namely the NHS and local government. This study provides evidence on the significance of the relationship between the AC, IA, and EA which all contribute to having sound CG arrangements, governance processes, and structures in public sector organisations. Maintaining sound accountability relationships within organisations with the various stakeholders involved is a major challenge faced by public sector organisations due to the complexities involved. Nevertheless, through the ACs challenging and monitoring responsibilities, one of which is overseeing the roles of the IA and EA, ACs have a significant role in making sure standards and accountability relationships enhanced.

The study has several contributions. First, it contributes to the claims that "researchers in accounting have not been responsive to the problems and opportunities associated with developments in public sector accounting" (Hay & Cordery, 2017; O'Leary, 2015; Broadbent & Guthrie, 2008). Secondly, it adds to the scant literature on

public sector auditing in a different organisational setting especially on the IA's effectiveness and IA-EA relationships (Postula et al., 2020; Kotb et al., 2020). Thirdly, it is one of the few studies in a public sector context that uses a qualitative study approach to gain insights from organisations through within semi-structured interviews and meeting observation the interaction between the AC, IA, and EA in practice. Finally, from a theoretical perspective, it provides further evidence that the application of the theoretical assumptions of AT to the public sector is not sufficient and only partly provides explanations and that multiple theories can be simultaneously to understand the adoption of CG processes and structures in public sector organisations. Whilst the study provides evidence of having sound CG arrangements in line with agency theory characteristics, in terms of the IA and EA roles and their relationship with the AC, the relative newness of such arrangements in LAs provides some evidence of loosely coupled systems in terms of absorbing changes in a ceremonial way. Further research is to be done in the future to observe such developments and may be extended to other public sector institutional settings and organizational contexts such as in schools, universities, and central government. Such studies can also be applied and used at a national or an international level and may open the door for comparisons between public and private sector organizations or between different environments in the public sector itself.

The limitations associated with the case study approach apply to this study as critics may argue that as the findings are based on a specific context, setting and only four cases, the claimed contributions to knowledge may not pass the test of statistical generalisation; that is, they lack external validity. Moreover, as the findings are based on a specific context within space and time it cannot be guaranteed to be the same in the future for other environments and institutional settings. However, such findings can be analytically generalizable, meaning that the findings can be generalizable to similar kinds of circumstances and environments (Ryan, Scapens, & Theobald, 2002). As Scapens and Macintosh (1996) have noted, analysis interpretation of this study's findings may be differently produced by different perspectives. For this reason, the interpretations and analysis and the resulting conclusions may be speculative and may reflect the biases including that of the researcher. The study has sought to control for such biases including adopting a triangulation approach by using various sources of data including direct quotations from interviews complemented by documentary analysis and meeting observations.

REFERENCES

- 1. Abbott, L. J., Parker, S., & Peters, G. F. (2010). Serving two masters: The association between audit committee internal audit oversight and internal audit activities. *Accounting Horizons*, 24(1), 1-24. https://doi.org/10.2308/acch.2010.24.1.1
- 2. Adams, M. B. (1994). Agency theory and the internal audit. *Managerial Auditing Journal*, *9*(8), 8–12. https://doi.org/10.1108/02686909410071133
- 3. Ahrens, T., & Chapman, C. S. (2006). Theorizing practice in management accounting research. *Handbooks of Management Accounting Research*, 1, 99-112. https://doi.org/10.1016/S1751-3243(06)01003-0

- 4. Ahrens, T., & Dent, J. F. (1998). Accounting and organizations: Realizing the richness of field research. *Journal of Management Accounting Research*, 10, 1–39. Retrieved from https://cutt.ly/lnm3Ldq
- 5. Alzeban, A. (2015). Influence of audit committees on internal audit conformance with internal audit standards. *Managerial Auditing Journal*, 10(4), 26–34. https://doi.org/10.5539/ijbm.v10n4p26
- 6. Alzeban, A., & Gwilliam, D. (2014). Factors affecting the internal audit effectiveness: A survey of the Saudi public sector. *Journal of International Accounting, Auditing and Taxation, 23*(2), 74–86. https://doi.org/10.1016/j.intaccaudtax.2014.06.001
- 7. Aristovnik, A., & Seljak, J. (2009). *Performance budgeting: Selected international experiences and some lessons for Slovenia* (MPRA Paper No. 15499). Retrieved from https://mpra.ub.uni-muenchen.de/15499/1/MPRA_paper_15499.pdf
- 8. Asiedu, K. F., & Deffor, E. W. (2017). Fighting corruption by means of effective internal audit function: Evidence from the Ghanaian public sector. *International Journal of Auditing*, 21(1), 82–99. https://doi.org/10.1111/ijau.12082
- 9. Audit Commission. (2010). *Code of audit practice 2010: Local government bodies*. Retrieved from https://www.leicestershireandrutlandalc.gov.uk/uploads/lg-code-of-audit-practice.pdf
- 10. Ball, A., Grubnic, S., & Birchall, S. J. (2014). Sustainability accounting and accountability in the public sector. In J. Bebbington, J. Unerman, & B. O'Dwyer (Eds.), *Sustainability accounting and accountability* (p. 176–196). https://doi.org/10.4324/9781315848419
- 11. Beasley, M. S., Carcello, J. V., Hermansson, D. R., & Neal, T. L. (2009). The audit committee oversight process. *Contemporary Accounting Research*, 26(1), 65–122. https://doi.org/10.1506/car.26.1.3
- 12. Bédard, J., & Gendron, Y. (2010). Strengthening the financial reporting system: Can audit committees deliver? *International Journal of Auditing*, 14(2), 174–210. https://doi.org/10.1111/j.1099-1123.2009.00413.x
- 13. Blue Ribbon Committee. (1999). Report and recommendations of the Blue Ribbon Committee on improving the effectiveness of corporate audit committees. *The Business Lawyer*, *54*(3), 1067–1095. Retrieved from https://www.jstor.org/stable/40687877
- 14. Bovens, M. (2007). Analysing and assessing accountability: A conceptual framework. *European Law Journal,* 13(4), 447-468. https://doi.org/10.1111/j.1468-0386.2007.00378.x
- 15. Bovens, M. (2010). Two concepts of accountability: Accountability as a virtue and as a mechanism. *West European Politics*, 33(5), 946–967. https://doi.org/10.1080/01402382.2010.486119
- 16. Bradbury, M., & Scott, T. (2015). The association between accounting performance and constituent response in political markets. *Pacific Accounting Review*, *27*(4), 394-410. https://doi.org/10.1108/PAR-02-2014-0007
- 17. Brierley, J. A., El-Nafabi, H. M., & Gwilliam, D. R. (2001). The problems of establishing internal audit in the Sudanese public sector. *International Journal of Auditing*, 5(1), 73–87. https://doi.org/10.1111/1099-1123.00326
- 18. Broadbent, J., & Guthrie, J. (2008). Public sector to public services: 20 years of "contextual" accounting research. *Accounting, Auditing & Accountability Journal*, 21(2), 129–169. https://doi.org/10.1108/09513570810854383
- 19. Burnett, C., Norman, T. J., & Sycara, K. (2011). Trust decision-making in multi-agent systems. *Proceedings of the Twenty-Second International Joint Conference on Artificial Intelligence* (pp. 115–120). http://doi.org/10.5591/978-1-57735-516-8/IJCAI11-031
- 20. Carcello, J. V., & Neal, T. L. (2000). Audit committee composition and auditor reporting. *The Accounting Review,* 75(4), 453–467. https://doi.org/10.2308/accr.2000.75.4.453
- 21. Carcello, J. V., & Neal, T. L. (2003). Audit committee characteristics and auditor dismissals following "new" going-concern reports. *The Accounting Review, 78*(1), 95–117. https://doi.org/10.2308/accr.2003.78.1.95
- 22. Carcello, J. V., Eulerich, M., Masli, A., & Wood, D. A. (2020). Are internal audits associated with reductions in perceived risk? *Auditing: A Journal of Practice & Theory, 39*(3), 55–73. https://doi.org/10.2308/ajpt-19-036
- 23. Chartered Institute of Internal Auditors. (2020). *Internal audit code of practice: Guidance on effective internal audit in the private and third sector*. Retrieved from https://www.iia.org.uk/media/1691066/internal-audit-code-of-practice-report.pdf
- 24. Chen, K. Y., & Zhou, J. (2007). Audit committee, board characteristics, and auditor switch decisions by Andersen's clients. *Contemporary Accounting Research*, *24*(4), 1085–1117. https://doi.org/10.1506/car.24.4.2
- 25. Christopher, J. (2010). Corporate governance A multi-theoretical approach to recognizing the wider influencing forces impacting on organizations. *Critical Perspectives on Accounting*, *21*(8), 683–695. https://doi.org/10.1016/j.cpa.2010.05.002
- 26. Coetzee, S. A., van Rensburg, C. J., & Schmulian, A. (2016). Differences in students' reading comprehension of international financial reporting standards: A South African case. *Accounting Education*, *25*(4), 306–326. https://doi.org/10.1080/09639284.2016.1191269
- 27. Cohen, J., & Krishnamoorthy, G., & Wright, A. M. (2008). Form versus substance: The implication for auditing practice and research of alternative perspectives on corporate governance. *AUDITING: A Journal of Practice & Theory, 27*(2), 181–198. https://doi.org/10.2308/aud.2008.27.2.181
- 28. Cohen, J., Krishnamoorthy, G., & Wright, A. M. (2007). Corporate governance and the audit process. *Contemporary Accounting Research*, 19(4), 573–594. https://doi.org/10.1506/983M-EPXG-4Y0R-J9YK
- 29. Cohen, J., Krishnamoorthy, G., & Wright, A. M. (2010). Corporate governance in the post-Sarbanes-Oxley era: Auditors' experiences. *Contemporary Accounting Research*, *27*(3), 751–786. https://doi.org/10.1111/j.1911-3846.2010.01026.x
- 30. Compernolle, T. (2018). Communication of the external auditor with the audit committee: Managing impressions to deal with multiple accountability relationships. *Accounting, Auditing & Accountability Journal,* 31(3), 900–924. https://doi.org/10.1108/AAAJ-05-2013-1356
- 31. Crilly, D., Zollo, M., & Hansen, M. T. (2012). Faking it or muddling through? Understanding decoupling in response to stakeholder pressures. *Academy of Management Journal*, 55(6), 1429–1448. https://doi.org/10.5465/amj.2010.0697
- 32. Davies, M. (2009). Effective working relationships between audit committees and internal audit The cornerstone of corporate governance in local authorities, a Welsh perspective. *Journal of Management & Governance*, 13(1-2), 41–73. https://doi.org/10.1007/s10997-008-9070-9

- 33. Department of Health. (2011).The NHS internal audit standards. https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/212826/ NHS-internal-audit-standards-april-2011.pdf
- 34. Eulerich, M., Kremin, J., & Wood, D. A. (2019). Factors that influence the perceived use of the internal audit function's work by executive management and audit committee. Advances in Accounting, 45, 100410. https://doi.org/10.1016/j.adiac.2019.01.001
- 35. Ezzamel, M., Robson, K., Stapleton, P., & McLean, C. (2007). Discourse and institutional change: 'Giving accountability. accounts' Management Accounting Research, 18(2), 150-171. and https://doi.org/10.1016/j.mar.2007.03.001
- 36. Financial Reporting Council (FRC). (2003). Audit committee combined code guidance. Retrieved from https://cutt.ly/CnEiOKe
- Financial Reporting Council (FRC). (2018). The UK corporate governance code. Retrieved from https://www.frc.org.uk/getattachment/88bd8c45-50ea-4841-95b0-d2f4f48069a2/2018-UK-Corporate-Governance-Code-FINAL.pdf
- Freeman, R., & Evan, W. M. (1990). Corporate governance: A stakeholder interpretation. Journal of Behavioural Economics, 19(4), 337–359. https://doi.org/10.1016/0090-5720(90)90022-Y
- 39. Gailmard, S. (2014). Accountability and principal Agent theory. In M. Bovens, R. E. Goodin, & T. Schillemans (Eds.), The Oxford handbook of public accountability. https://doi.org/10.1093/oxfordhb/9780199641253.013.0016
- 40. Gendron, Y. (2009). Discussion of "The audit committee oversight process": Advocating openness in accounting research. Contemporary Accounting Research, 26(1), 123-134. https://doi.org/10.1506/car.26.1.4
- Gendron, Y., & Bédard, J., (2006). On the constitution of audit committee effectiveness. *Accounting, Organizations and Society*, *31*(3), 211–239. https://doi.org/10.1016/j.aos.2005.03.002
- Gendron, Y., Bédard, J., & Gosselin, M. (2004). Getting inside the black box: A field study of practices in "effective" audit committees. Auditing: A Journal of Practice & Theory. *23*(1). https://doi.org/10.2308/aud.2004.23.1.153
- 43. Getie Mihret, D., & Wondim Yismaw, A. (2007). Internal audit effectiveness: An Ethiopian public sector case study. *Managerial Auditing Journal*, 22(5), 470–484. https://doi.org/10.1108/02686900710750757 Glynn, J. J. (1985). Value for money auditing — An international review and comparison, *Financial*
- 44. Glynn, J. J. (1985). Value for money auditing Accountability and Management, 1(2), 113-128. https://doi.org/10.1111/j.1468-0408.1985.tb00248.x
- 45. Goddard, A. (2010). Contemporary public sector accounting research An international comparison of journal papers. The British Accounting Review, 42(2), 75-87. https://doi.org/10.1016/j.bar.2010.02.006
- Goodwin, J. (2004). A comparison of internal audit in the private and public sectors. Managerial Auditing Journal, 19(5), 640-650. https://doi.org/10.1108/02686900410537766
- 47. Gramling, A. A., Maletta, M. J., Schneider, A., & Church, B. K. (2004). The role of the internal audit function in corporate governance: A synthesis of the extant internal auditing literature and directions for future research. Journal of Accounting Literature, 23, 194-244. Retrieved from https://cutt.ly/DnWw1qm
- Greenwood, R., Raynard, M., Kodeih, F., Micelotta, E. R., & Lounsbury, M. (2011). Institutional complexity and organizational responses. Academy Management Annals. 5(1). of https://doi.org/10.5465/19416520.2011.590299
- 49. Griffee, D. T. (2012). An introduction to second language research methods. Retrieved from http://www.teslej.org/pdf/ej60/sl_research_methods.pdf
- 50. Hay, D., & Cordery, C. (2018). The value of public sector audit: Literature and history. Journal of Accounting Literature, 40, 1-15. https://doi.org/10.1016/j.acclit.2017.11.001
- Higgs, D. (2003). Review of the role and effectiveness of non-executive directors. Retrieved from https://ecgi.global/sites/default/files//codes/documents/higgsreport.pdf
- HM Treasury. (2011). Corporate governance in central government departments: Code of good practice 2011. https://assets.publishing.service.gov.uk/government/uploads/system/uploads/ from attachment_data/file/220645/corporate_governance_good_practice_july2011.pdf
- HM Treasury. (2013). Managing public money. Retrieved from www.gov.uk/government/publications/managingpublic-money
- A public management for all seasons? Public administration, https://doi.org/10.1111/j.1467-9299.1991.tb00779.x
- Hood, C. (1995). The "new public management" in the 1980s: Variations on a theme. Accounting, Organizations, and Society, 20(2-3), 93-109. https://doi.org/10.1016/0361-3682(93)E0001-W
- 56. Hudaib, M., & Cooke, T. E. (2005). The impact of managing director changes and financial distress on audit qualification and auditor switching. *Journal of Business Finance & Accounting*, 32(9–10), 1703–1739. https://doi.org/10.1111/j.0306-686X.2005.00645.x
- 57. Jacobs, K. (2012). Making sense of social practice: Theoretical pluralism in public sector accounting research. Financial Accountability & Management, 28(1), 1-25. https://doi.org/10.1111/j.1468-0408.2011.00534.x
- 58. Jarrar, Y., & Schiuma, G. (2007). Measuring performance in the public sector: Challenges and trends. *Measuring* Business Excellence, 11(4), 4-8. https://doi.org/10.1108/13683040710837883
- Kotb, A., Elbardan, H., & Halabi, H. (2020). Mapping of internal audit research: A post-Enron structured Auditing literature review. Accounting, Accountability Journal, 33(8), https://doi.org/10.1108/AAAJ-07-2018-3581
- Krishnan, J. (2005). Audit committee quality and internal control: An empirical analysis. The Accounting Review, 80(2), 649-675. https://doi.org/10.2308/accr.2005.80.2.649
- 61. Lan, L. L., & Heracleous, L. (2010). Rethinking agency theory: The view from law. Academy of Management Review, 35(2), 294–314. https://doi.org/10.5465/amr.35.2.zok294
- 62. Lenz, R., & Hahn, U. (2015). A synthesis of empirical internal audit effectiveness literature pointing to new research opportunities. Managerial Auditing Journal, 30(1), 5-33. https://doi.org/10.1108/MAJ-08-2014-1072
- Lok, J. (2010). Institutional logics as identity projects. Academy of Management Journal, 53(6), 1305-1335. https://doi.org/10.5465/amj.2010.57317866
- 64. Marriott, N., Mellett, H., & Macniven, L. (2011). Loose coupling in asset management systems in the NHS. Management Accounting Research, 22(3), 198-208. https://doi.org/10.1016/j.mar.2011.05.001

- 65. McKernan, J. (2013). Curriculum action research: A handbook of methods and resources for the reflective practitioner. https://doi.org/10.4324/9781315041742
- 66. Meier, K. J., & Krause, G. A. (Eds.). (2003). The scientific study of bureaucracy: An overview. In *Politics, policy, and organizations: Frontiers in the scientific study of bureaucracy* (1–19). Retrieved from https://cutt.ly/FnWt8xc
- 67. Meyer, J. W., & Rowan, B. (1977). Institutionalized organizations: Formal structure as myth and ceremony. *American Journal of Sociology*, *83*(2), 340–363. https://doi.org/10.1086/226550
- 68. Modell, S. (2009a). In defence of triangulation: A critical realist approach to mixed methods research in management accounting. *Management Accounting Research*, 20(3), 208–221. https://doi.org/10.1016/j.mar.2009.04.001
- 69. Modell, S. (2009b). Institutional research on performance measurement and management in the public sector accounting literature: A review and assessment. *Financial Accountability & Management, 25*(3), 277–303. https://doi.org/10.1111/j.1468-0408.2009.00477.x
- 70. Monitor. (2013). *Your duties: A brief guide for NHS foundation trust governors.* Retrieved from https://www.bcpft.nhs.uk/documents/about-us/assembly-of-governors/2105-brief-guide-for-governors/file
- 71. Monitor. (2014). *The NHS foundation trusts code of governance*. Retrieved from https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/327068/CodeofGovernanceJuly2014.pdf
- 72. National Audit Office (NAO), & Office of Government Commerce (OGC). (2000). *Getting value for money from procurement: How auditors can help?* Retrieved from https://www.nao.org.uk/report/getting-value-for-money-from-procurement-how-auditors-can-help/
- 73. Nor-Aziah, A. K., & Scapens, R. W. (2007). Corporatisation and accounting change: The role of accounting and accountants in a Malaysian public utility. *Management Accounting Research*, 18(2), 209–247. https://doi.org/10.1016/j.mar.2007.03.003
- 74. Office for National Statistics (ONS). (2021). *UK economy latest.* Retrieved from https://www.ons.gov.uk/economy/economicoutputandproductivity/output/articles/ukeconomylatest/2021-01-25
- 75. O'Leary, C., & Stewart, J. (2007). Governance factors affecting internal auditors' ethical decision-making: An exploratory study. *Managerial Auditing Journal*, 22(8), 787–808. https://doi.org/10.1108/02686900710819643
- 76. O'Leary, D. E. (2015). Crowdsourcing tags in accounting and finance: Review, analysis, and emerging issues. *Journal of Emerging Technologies in Accounting*, *12*(1), 93–115. https://doi.org/10.2308/jeta-51195
- 77. Oliver, C. (1991). Strategic responses to institutional processes. *Academy of Management Review*, 16(1), 145-179. https://doi.org/10.5465/amr.1991.4279002
- 78. Parker, L. D., & Guthrie, J. (1993). The Australian public sector in the 1990s: New accountability regimes in motion. *Journal of International Accounting, Auditing and Taxation, 2*(1), 59–81. https://doi.org/10.1016/1061-9518(93)90015-L
- 79. Patton, M. Q. (2002). Two decades of developments in qualitative inquiry: A personal, experiential perspective. *Qualitative Social Work*, 1(3), 261–283. https://doi.org/10.1177/1473325002001003636
- 80. Postula, M., Irodenko, O., & Dubel, P. (2020). Internal audit as a tool to improve the efficiency of public service. *European Research Studies*, *23*(3), 699–715. https://doi.org/10.35808/ersj/1663
- 81. Privett, N., & Erhun, F. (2011). Efficient funding: Auditing in the nonprofit sector. *Manufacturing & Service Operations Management*, 13(4), 471–488. https://doi.org/10.1287/msom.1110.0345
- 82. Rezaee, Z., & Lander, G. H. (1993). The internal auditor's relationship with the audit committee. *Managerial Auditing Journal*, 8(3). https://doi.org/10.1108/02686909310036269
- 83. Roussy, M. (2013). Internal auditors' roles: From watchdogs to helpers and protectors of the top manager. *Critical Perspectives on Accounting*, 24(7–8), 550–571. https://doi.org/10.1016/j.cpa.2013.08.004
- 84. Ryan, B., Scapens, R. W., & Theobald, M. (2002). *Research methods and methodology in accounting and finance* (2nd ed.). London, UK: Thompson Learning.
- 85. Sarens, G., De Beelde, I., & Everaert, P. (2009). Internal audit: A comfort provider to the audit committee. *The British Accounting Review*, 41(2), 90-106. https://doi.org/10.1016/j.bar.2009.02.002
- 86. Scapens, R. W., & Macintosh, N. B. (1996). Structure and agency in management accounting research: A response to Boland's interpretive act. *Accounting, Organizations and Society, 21*(7–8), 675–690. https://doi.org/10.1016/0361-3682(95)00049-6
- 87. Schillemans, T. (2013). Moving beyond the clash of interests: On stewardship theory and the relationships between central government departments and public agencies. *Public Management Review*, *15*(4), 541–562. https://doi.org/10.1080/14719037.2012.691008
- 88. Schillemans, T., & Busuioc, M. (2015). Predicting public sector accountability: From agency drift to forum drift. *Journal of Public Administration Research and Theory*, 25(1), 191–215. https://doi.org/10.1093/jopart/muu024
- 89. Spira, L. F. (2002). The audit committee: Performing corporate governance. https://doi.org/10.1007/b110877
- 90. Spira, L. F. (2003). Audit committees: Begging the question? *Corporate Governance: An International Review,* 11(3), 180–188. https://doi.org/10.1111/1467-8683.00317
- 91. Stewart, J., & Subramaniam, N. (2010). Internal audit independence and objectivity: Emerging research opportunities. *Managerial Auditing Journal*, 25(4), 328–360. https://doi.org/10.1108/02686901011034162
- 92. The Chartered Institute of Public Finance & Accountancy (CIPFA), Department of Health, Llywodraeth Cymru Welsh Government, Department of Finance An Roinn Airgeadais, HM Treasury, Scottish Government Riaghaltas na h-Alba, & Chartered Institute of Internal Auditors. (2017). Public sector internal audit standards: Applying the IIA international standards to the UK public sector. Retrieved from https://www.gov.uk/government/publications/public-sector-internal-audit-standards
- 93. The Chartered Institute of Public Finance and Accountancy (CIPFA). (2013). *Audit committees: Practical guidance for local authorities*. London, UK: CIPFA.
- 94. The Sarbanes-Oxley Act of 2002, Public Law 107–204, July 30, 2002. Retrieved from https://pcaobus.org/About/History/Documents/PDFs/Sarbanes_Oxley_Act_of_2002.pdf
- 95. Turley, S., & Zaman, M. (2007). Audit committee effectiveness: Informal processes and behavioural effect. *Accounting, Auditing & Accountability Journal*, 20(5), 765–788. https://doi.org/10.1108/09513570710779036

- 96. Van Gansberghe, C.N. (2005). Internal auditing in the public sector: A consultative forum in Nairobi, Kenya, shores up best practices for government audit professionals in developing nations. *Internal Auditor*, *62*(4), 69–74. Retrieved from https://cutt.ly/2nUbUx2
- 97. Van Schendelen, R. (2012). New trends of public affairs management at the EU level. *Journal of Public Affairs*, 12(1), 39–46. https://doi.org/10.1002/pa.380
- 98. Weick, K. E. (1976). Educational organizations as loosely coupled systems. *Administrative Science Quarterly*, 21(1), 1-19. https://doi.org/10.2307/2391875
- 99. Yin, R. K. (Ed.). (2003). Case study research: Design and methods. Thousand Oaks, CA: SAGE Publications.
- 100. Zain, M. M., & Subramaniam, N. (2007). Internal auditor perceptions on audit committee interactions: A qualitative study in Malaysian public corporations. *Corporate Governance: An International Review, 15*(5), 894–908. https://doi.org/10.1111/j.1467-8683.2007.00620.x
- 101. Zaman, M., & Sarens, G. (2013). Informal interaction between audit committees and internal audit functions: Exploratory evidence and directions for future research. *Managerial Auditing Journal*, *28*(6), 495–515. https://doi.org/10.1108/02686901311329892
- 102. Zhang, Y., Zhou, J., & Zhou, N. (2007). Audit committee quality, auditor independence and internal control weaknesses. *Journal of Accounting and Public Policy, 26*(3), 300–327. https://doi.org/10.1016/j.jaccpubpol.2007.03.001